Author's response to reviews

Title: Primary extracranial meningioma in submandibular region - A case report

Authors:

Sanjay D Deshmukh (drsanjay123in@yahoo.co.in)
Vidya V Rokade (vidya_rokade@rediffmail.com)
Gayatri S Pathak (drgayatridp@yahoo.in)
Sanjana V Nemade (drsanjana31@yahoo.in)
Amrut V Ashturkar (amrutva@yahoo.com)

Version: 6 Date: 31 March 2011

Author's response to reviews: see over
Covering letter

To the Editorial board,
JMCR

Subject : MS: 8986022044240133- Primary extracranial meningioma in submandibular region - A case report

Many thanks for your communication.

As per your suggestion we have revised the manuscript. The changes in this revision are highlighted in green color.

1) We have revised figure legends on page number 10.

2) Corresponding figure number has been added in the text. Following changes made-
   Figure 1 – Page 3 line 13
   Figure 2 – Page 3 line 16
   Figure 3 – Page 4 line 2
   Figure 4 – Page 4 line 15
   Figure 5 – Page 4 line 22

3) As per reviewer’s comment arrows has been added in figure 2, figure 3 and figure 4.

Thanking you in anticipation.

Dr Sanjay D Deshmukh
Professor of Pathology.
Shrimati Kashibai Navale Medical College and General Hospital, Narhe, Pune-411041, India